Author's response to reviews

**Title:** Up-regulated expression of type II very low density lipoprotein receptor correlates with cancer metastasis and has a potential link to beta-catenin in different cancers

**Authors:**

Lei He (atoprine@qq.com)  
Yanjun Lu (junyanlu_2000@163.com)  
Peng Wang (wpcz_1003@163.com)  
Jun Zhang (zhangjun1008@126.com)  
Chuanchang Yin (yinchuanchang8260464@126.com)  
Shen Qu (qs_cn@sina.com)

**Version:** 6  **Date:** 1 November 2010

**Author's response to reviews:** see over
Dear editors:

It gives us great pleasure to learn that our manuscript is acceptable for publication in BMC Cancer, and many thanks for the approval to our work. We have checked the manuscript carefully, and listed some changes below. We greatly appreciate both your help and that of the reviewers concerning improvement to this paper. We hope that the revised manuscript is now suitable for publication. Please contact us if there are any problems or questions.

Best wishes,

Yours Sincerely
Shen Qu
November 1, 2010

Major revisions

1) We removed the wrongly repeated sentence “would be assigned the maximum” in page 8, line 15 of last version (version 5).

2) The statement about source of foundation “This work was supported by grants from the National Natural Science Foundation of China (No. 30470872, 30300134) and the Doctor Fund Project of Chinese Ministry of Education (No. 20090142110014)” had been added into the part of Acknowledgements of the revised manuscript (page 19, line 11).

Minor revisions

3) We have corrected some spelling or grammar mistakes, and adjusted the figures according to your online figure guide.